Author's response to reviews

Title: COX-2, CB2 and P2X7-immunoreactivities are increased in activated microglial cells/macrophages of multiple sclerosis and amyotrophic lateral sclerosis spinal cord

Authors:
Yiangos Yiangou (Y.Yiangou@imperial.ac.uk)
Pascal Durrenberger (P.Durrenberger@imperial.ac.uk)
Iain P Chessell (lain.P.Chessell@gsk.com)
Richard R Banati (r.banati@fhs.usyd.edu.au)

Version: 5 Date: 13 February 2006

Author's response to reviews:

Please find attached a final version with formatting changes and minor spelling corrections that you have requested. Please also note that endnote keeps adding an 'et al.' after ALL the authors in reference 2 and this has to be removed manually.